Course Information Sheet, CSC238 session L0101, Fall 2002

This sheet contains most of the important information you need to know about this course. For more detailed information see the course website:

http://www.cs.toronto.edu/~mreza/CSC238

Note: You must read the “Main page” of this website as it contains some important information regarding the course. During the session, announcements will be made at the begining of lectures and will be posted on this website. It is your responsibility to check this site frequently (at least twice a week).

Instructor Information:

ffl Instructor: Mohammad R. Salavatipour ffl E-mail: mreza@cs.toronto.edu ffl Office: SF3208 ffl Phone: (416)978-8700

The best way to contact me is by e-mail. The phone line is shared with my officemates. So PLEASE call me only if it is an emergency.

Time tables:

ffl Lecture: MF 10-11, in MB 128.

ffl Tutorial: W 10-11, in MB 128, RS 208, RS 310, GB 304, and GB 404.

Your tutorial section will be announced soon, on the course website. ffl Office hour: M 3-4, TBA.

Text book:

The text book for this course is: “Course notes for CSC238/B38: DISCRETE MATHEMAT- ICS”, by Vassos Hadzilacos, 2002-2003 edition. See the website for additional references.

Grading Scheme:

ffl Five assignments at 6% each: 30%. ffl Two term tests at 15% each: 30%. ffl Final exam: 40%.

NOTE:

ffl In order to pass the course, you must achieve at least 40% on your final exam, regardless of your other marks.

ffl For any problem in an assignment or term test (but not for the final exam), you will receive 20% of the mark for that problem if you only write "I don’t know how to answer to this question", and nothing else!!.

Important Dates:

Date Event Mon, Sep 9 First lecture Wed, Sep 11 First tutorial (Check announcements on the website) Wed, Sep 18 Assignment 1 posted Wed, Oct 2 Assignment 1 due, Assignment 2 posted Wed, Oct 9 Term test 1 Mon, Oct 14 Thanksgiving, No lecture Wed, Oct 16 Assignment 2 due, Assignment 3 posted. No tutorials; lecture in MB 128 at the same time.

Wed, Oct 30 Assignment 3 due, Assignment 4 posted Sun, Nov 3 Last day to drop the course Wed, Nov 6 Term test 2 Wed, Nov 13 Assignment 4 due, Assignment 5 posted Wed, Nov 27 Assignment 5 due Fri, Dec 6 Last lecture

Assignments and Plagiarism:

All assignments are due at the begining of tutorial. Late assignments are NOT accepted!!. If you think there is a chance that you cannot come to your tutorial to submit your assignment, submit it earlier to either your instructor or your TA.

Your assignments must not contain anyone else’s ideas. That is, you are not allowed to discuss the assignments with ANYBODY (other than your instructor and your TA). In particular, you must write your assignments isolated from your friends and other students. Do NOT let other students see your solutions, since this is also considered as plagiarism. We treat plagiarism very seriously. See the website for more information.